# Common alleles of *CMT2* and *NRPE1* are major determinants of *de novo* DNA methylation variation in *Arabidopsis thaliana*

Eriko Sasaki<sup>1</sup>, Taiji Kawakatsu<sup>2,3,5</sup>, Joseph Ecker<sup>2,3,4</sup>, Magnus Nordborg<sup>1,\*</sup>

 Gregor Mendel Institute of Molecular Plant Biology, Austrian Academy of Sciences, Vienna Biocenter, 1030 Vienna, Austria
 Plant Biology Laboratory
 Genomic Analysis Laboratory

4 Howard Hughes Medical Institute

The Salk Institute for Biological Studies, La Jolla, CA 92037, USA

5 Institute of Agrobiological Sciences, National Agriculture and Food Research Organization. Tsukuba, Ibaraki 305-8634, Japan

\* magnus.nordborg@gmi.oeaw.ac.at

# Abstract

DNA cytosine methylation is an epigenetic mark associated with silencing of transposable elements (TEs) and heterochromatin formation. In plants, it occurs in three sequence contexts: CG, CHG, and CHH (where H is A, T, or C). The latter does not allow direct inheritance of methylation during DNA replication due to lack of symmetry, and methylation must therefore be re-established every cell generation. Genome-wide association studies (GWAS) have previously shown that CMT2 and NRPE1 are major determinants of genome-wide patterns of TE CHH-methylation. Here we instead focus on CHH-methylation of individual TEs and TE-families, allowing us to identify the pathways involved in CHH-methylation simply from natural variation and confirm the associations by comparing them with mutant phenotypes. Methylation at TEs targeted by the RNA-directed DNA methylation (RdDM) pathway is unaffected by CMT2 variation, but is strongly affected by variation at NRPE1, which is largely responsible for the longitudinal cline in this phenotype. In contrast, CMT2-targeted TEs are affected by both loci, which jointly explain 7.3% of the phenotypic variation (13.2% of total genetic effects). There is no longitudinal pattern for this phenotype, however, because the geographic patterns appear to compensate for each other in a pattern suggestive of stabilizing selection.

# Author Summary

DNA methylation is a major component of transposon silencing, and essential for genomic integrity. Recent studies revealed large-scale geographic variation as well as the existence of major *trans*-acting polymorphisms that partly explained this variation. In this study, we re-analyze previously published data (The 1001 Epigenomes), focusing on *de novo* DNA methylation patterns of individual TEs and TE families rather than on genome-wide averages (as was done in previous studies). GWAS of the patterns reveals the underlying regulatory networks, and allowed us to comprehensively characterize

*trans*-regulation of *de novo* DNA methylation and its role in the striking geographic pattern for this phenotype.

# Introduction

DNA cytosine-methylation (DNA methylation) is an epigenetic mark associated with diverse molecular functions, such as silencing of transposable elements (TEs) and heterochromatin formation. The majority of plant methylation is found in TEs, and there are three types of DNA methylation contexts: CG and CHG, both of which are symmetric, and CHH, which is not (H is A, T, or C). CG-methylation (mCG) and CHG-methylation (mCHG) can be maintained in a semi-conservative manner during DNA replication by DNA METHYLTRANSFERASE 1 (MET1) and CHROMOMETHYLASE 3 (CMT3), respectively, whereas CHH-methylation (mCHH) must be re-established every cell generation, presumably by one of two de novo pathways, one involving CHROMOMETHYLASE 2 (CMT2), the other RNA-directed DNA methylation (RdDM) [1–3]. CMT2 preferentially methylates heterochromatic non-CG cytosines that are marked by H3 Lys9 (H3K9) di- and tri-methylation [4,5], whiles RdDM involves small RNAs that recruit DOMAINS REARRANGED METHYLTRANSFERASE 2 (DRM2) to target regions throughout the genome [6, 7]. These pathways thus have separate target sites [4] and establish the genome-wide DNA methylation landscape in combination with maintenance and de-methylation pathways.

Natural variation for DNA methylation, superficially similar to DNA sequence polymorphism, is abundant in Arabidopsis [8,9]. Although much of this variation likely reflects local sequence variation (e.g. segregating TE insertions), recent studies have revealed that a substantial part of the variation is controlled by *trans*-acting loci with genome-wide effects [10–12]. Understanding these *trans*-regulators is essential for understanding the genome-wide pattern of methylation variation, and could provide important clues to the function of DNA methylation.

The present study builds on previous results to comprehensively characterize *trans*-regulation of mCHH and its role in the striking geographic pattern for this phenotype. We achieve this by looking for genotype-phenotype associations at the level of individual TEs or TE families rather than genome-wide averages. As we shall see, this makes a huge difference.

# Results

# Major trans-regulators of mCHH levels

We first characterized average mCHH profiles of 303 TE families in each individual (S1 Tablel). Clustering analysis (based on the pattern across the 774 individuals) identified four groups, with the largest two roughly corresponding to the TE families that were previously shown to lose mCHH in RdDM and CMT2 pathway mutants (Fig. 1 [13]). The group corresponding to the RdDM pathway is mostly class I TEs and is enriched with RC/Helitron and DNA/MuDR, whereas the group corresponding to the CMT2 pathway is dominated by class II TEs and is enriched with LTR/Copia and LTR/Gypsy (note that targeting also strongly depends on element length and genome location; see S2 Fig.)

GWAS for average mCHH levels of each TE family also identified the two main groups. Of 13 significant peaks (at FDR 20% and taking linkage disequilibrium (LD) into account; see Methods; Fig 1 and S2-3 Tables), six are associated with the group corresponding to the RdDM pathway with strong signals at chr2:16719071 in the coding region of NUCLEAR RNA POLYMERASE D1B (NRPE1) as the largest component of

1

g

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

30

31

32

33

34

35

36

37

38

39

40

41

42

43

RNA-polymerase V responsible for the RdDM pathway [7] and at chr1:17895231 in the promoter region of ARGONAUTE1 (AGO1) recruits 21nt small RNA [14]. The remaining seven peaks are associated with the group corresponding to the CMT2 pathway with very strong signals at chr4:10417744 and chr4:10422486 in the coding or 3' region of CHROMOMETHYLASE2 (CMT2) [5] (Fig. 1).

The pattern of natural variation in mCHH is thus sufficient to outline pathways previously painstakingly discovered using traditional genetic screens, as well as to identify some of the major genes involved.

In addition to known genes, nine clear peaks suggest undescribed regulators of DNA methylation (S2-3 Tables). For example, the peak at chr1:27261944 is in the promoter region of a gene coding a DNAJ domain (At1g72416) that is a common component of DNA methylation reader comlex [15], and the peak at chr4:9595111 is upstream of a histone H3K4-specific methyltransferase SET7/9 family gene (At4g17080) implicated in histone modification.

The four peaks that correspond to obvious *a priori* candidates are consistent with previous results [11, 12]. The peak near AGO1 identifies the same top SNP as Kawakatsu et al [12, ] while the remaining three are in strong LD, but are much closer to the respective candidate genes, presumably because the present analysis, focusing on TE families rather than on average methylation levels, has higher resolution (Fig. 2A). Thus chr2:16719071 is in LD with the previously identified chr2:16724013 [12], but is in the coding region of NRPE1, where it is LD with 12 non-synonymous polymorphisms and a three bp in-frame indel in the RNA polymerase domain. Similarly, chr4:10417744 is in LD with chr4:10454628 CMT2b (see [11]), but inside the coding region of CMT2 and tagging two non-synonymous SNPs in the DNA methylase domain as well as a twelve base-pair deletion in the first exon. Finally, chr4:10422486 is in LD with chr4:10459127 CMT2a (see [11]), which is still outside the coding region, but presumably in the regulatory region.

For clarity, we will refer to the newly identified associations as NRPE1', CMT2b' and CMT2a'. The non-reference NRPE1' allele is associated with decreased mCHH levels, whereas the non-reference alleles of CMT2b' and CMT2a' have negative and positive effects, respectively, in agreement with previous results [11].

GWAS for mCHH levels of 9,228 individual TEs that are present in all 774 lines showed a very similar pattern to GWAS for individual TE families. Although the AGO1peak was much weaker, the signals at NRPE1', CMT2b', and CMT2a' remain strong even at the level of individual TEs, with NRPE1' explaining 6.6% of the average mCHH variation on RdDM-targeted TEs, whereas the two CMT2 alleles each explain about 4% (total 6.4%) of the variation on CMT2-targeted TEs (Fig. 2B). Because the effect sizes are so large, and because the genes target different chromosomal regions (NRPE1mainly affects TEs in chromosome arms, whereas CMT2 targets TEs in pericentromeric regions; see Fig. 2B), these polymorphisms contribute substantially to shaping the genome-wide landscape of mCHH levels (Fig. 2C), and the remainder of this paper will focus on them. bioRxiv preprint doi: https://doi.org/10.1101/819516; this version posted October 25, 2019. The copyright holder for this preprint (which was not certified by peer review) is the author/funder, who has granted bioRxiv a license to display the preprint in perpetuity. It is made available under a CC-BY 4.0 International license.



Figure 1. The genetics of mCHH methylation at the level of TE families. The heat map shows GWAS results for 303 TE families (each row is a family; columns indicate positions in genome; blue is more significant). The Manhattan plot on top shows integrated *p*-values from combining results across families (using  $X^2$ -statics). The horizontal line in the Manhattan plot gives FDR 20% threshold, with significant associations shown in yellow (see Methods, S1 Fig.). Arrows indicate previously identified associations [11, 12] also identified here. TE-families (rows) have been clustered based on average mCHH levels for 774 lines. The tip colors of the resulting tree correspond to TE superfamilies, and the superfamily composition for the four large clusters (Groups I-IV) is summarized by pie charts on the left. The greenish bars on the right show the reduction in mCHH levels of each TE family in *drm1 drm2* (RdDM pathway) and *cmt2* (CMT2 pathway) loss-of-function lines.

bioRxiv preprint doi: https://doi.org/10.1101/819516; this version posted October 25, 2019. The copyright holder for this preprint (which was not certified by peer review) is the author/funder, who has granted bioRxiv a license to display the preprint in perpetuity. It is made available under a CC-BY 4.0 International license.



Figure 2. NRPE1 and CMT2 are strong trans-regulators of mCHH levels. (A) Examples of zoomed-in Manhattan plots for individual TEs targeted by NRPE1 (AT3TE44975) and by CMT2 (AT1TE41860). Horizontal lines show the 5% Bonferroni-corrected p-value threshold. Rectangles show gene models for the alleles identified [16]. Red and black triangles on the protein domain models indicate nonsynonymous SNPs and indels. (B) GWAS results for mCHH levels of 9,228 individual TEs in 774 lines (heat map) in each row with the integrated p-values by  $X^2$ statics shown in the Manhattan plot above (yellow associations are significant using an FDR of 20%; see Methods). (C) Allelic effects on genome-wide mCHH levels (chromosome 5). Y-axis is the average differential mCHH levels between lines carrying alternative and reference alleles (300 Kbp sliding windows). The black arrow indicates the centromeric region.

#### Causality of NRPE1 and CMT2 alleles

Identifying the causal polymorphisms underlying a GWAS peak is notoriously difficult [17, 18]. However, because the phenotypes associated with the polymorphisms just described are so specific (multi-dimensional mCHH on hundreds or even thousands of specific TEs throughout the genome), it is possible to confirm the causal involvement of genes by comparison to mutant phenotypes. Specifically, we compared the estimated allelic effects of NRPE1', CMT2b', and CMT2a' on 9,228 TEs with the effects of knock-out mutations for 86 genes involved in gene-silencing, including NRPE1 and

93

94

95

CMT2 [13]. The correlation between natural allelic effects and knock-out mutation effects for these genes was high, with the specific TEs significantly affected by the NRPE1' allele in GWAS also being affected by the nrpe1-11 loss-of-function allele, and TEs significantly affected by the CMT2a' and CMT2b' alleles in GWAS also being affected by the cmt2 loss-of-function allele (Figs 3 and S3-S4).



Figure 3. The allelic effects of NRPE1', CMT2b', and CMT2a'. (A) Comparison of the alleles to loss-of-function mutations of the corresponding genes. Scatter plots show correlations of differential mCHH levels (DML) induced by alleles and mutants for each TE. DML for alleles was estimated as average differences of mCHH levels between lines carrying reference and non-reference alleles, whereas for mutants it was estimated between wild-type and the nrpe1-11 or cmt2 loss-of-function. Colors of dots in the scatter plots show the significance of the allelic effects as  $-\log_{10}p$ -value in GWAS. Density plots on Y and X-axis show distributions of the allelic effects for TEs. (B) Manhattan plots of *cis* peaks for *CMT2* expression (n=665; leaf tissue under 21°C) and effects of  $CMT_2$  alleles. Horizontal lines show the threshold (*p*-value 5% Bonferroni correction), and identified SNPs in meta-analysis for mCHH variation of TE families were labeled (FDR < 20%). Boxplot shows *CMT2* expression of lines carrying reference or CMT2a' alleles. \*\*\* indicates p-value < 0.01 (Welch's t-test).

Furthermore, the phenotypic correlation between CMT2b' and cmt2 was much 101 stronger than the correlation between CMT2b' and any other gene knockout (Fig. 4), 102 effectively confirming the causal role of CMT2 — the alternative explanation would be that the identified non-synonymous polymorphisms in  $CMT^2$  affect methylation via a closely linked unidentified gene that mimics the highly specific phenotypic effects of CMT2 much better than any of the 85 other analyzed genes in these well-studied pathways. The correlation between the effects of CMT2a' and cmt2 is notably weaker, perhaps because this allele affects expression like a moderate overexpressor (Figs 3B, 4). This may be worth exploring further. 109

96

97

98

99

bioRxiv preprint doi: https://doi.org/10.1101/819516; this version posted October 25, 2019. The copyright holder for this preprint (which was not certified by peer review) is the author/funder, who has granted bioRxiv a license to display the preprint in perpetuity. It is made available under aCC-BY 4.0 International license.



Figure 4. Comparison of the effects of mCHH variation associated with 13 natural alleles to variation induced by knocking out 86 different genes involved in DNA methylation. The heat map shows Spearman's correlation coefficients between SNP- and mutant-associated DMLs across 9,228 TEs. Both rows (mutants [13]) and columns (SNPs found in GWAS; see Fig. 1, S2-3 Tables) have been clustered by similarity in DML pattern.

*NRPE1*', by contrast, is clearly less specific, and showed strong correlations with loss-of-function phenotypes of nine genes in the RdDM pathway (including, of course, *NRPE1* itself). However, since none of these genes, nor any other plausible candidate, is located near *NRPE1* (S5 Fig), it seems reasonable to assume that the non-synonymous polymorphisms in this gene, particularly in RNA polymerase domain, cause a phenotype similar to knocking out *NRPE1* [19], rather than by somehow regulating an

unknown member of the RdDM pathway (S5-6 Figs). The relative lack of specificity of NRPE1 can also be seen from the comparison of natural alleles and knock-out mutations. Whereas variation at CMT2 affects only a subset of TEs, variation at NRPE1 affects all TEs, albeit to different extents (Fig. 3).

In summary, we feel confident that both the CMT2 and NRPE1 alleles involve cis-acting polymorphisms that affect the phenotype via the corresponding genes. How this is done is of course not clear, but we note again that both NRPE1' and CMT2b' are associated with multiple non-synonymous SNPs, and that CMT2a' is associated with increased CMT2 expression (Fig. 3). Note that the same analysis does not work for the AGO1 association, perhaps because the allelic effects are too small (S7 Fig).

#### Apparent higher target specificity for natural alleles

The natural alleles thus show similar patterns to knock-out mutants, albeit with some notable differences. CMT2b' preferentially affects the same TEs as cmt2 regardless of whether we consider the most significant or the largest effects (Figs 3-4, and S5). CMT2a' behaves similarly, but only when we consider the most significant effects, perhaps because this allele affects only CMT2 expression. NRPE1' is more interesting, because while it is similar to the knock-out mutation in not affecting the LTR/Gypsy superfamily, it clearly affects the RC/Helitron superfamily preferentially, whereas nrpe1-11 shows no such enrichment (S8A Fig).

This difference in specificity could be due to difference in target specificity between these alleles, but may also be explained by the population dynamics of TEs, because it turns out *nrpe1-11* strongly affects TE-superfamilies that have relatively low frequency in the population (like RathE3 cons and SINE; see S1,4 Tables and S8C Fig). These effects would be missed by the GWAS analysis of individual TEs, which only considers high-frequency insertions.

# NRPE1' allele broadly affects both the RdDM and CMT2-targeted regions

CMT2 and NRPE1 are considered to be parts of different pathways and target different TEs (Figs 1, 2). However, as noted above, variation at NRPE1 clearly affects methylation of CMT2-targeted TEs, whereas the converse is not true (Figs 3 and S9; p-value < 0.01).

We examined the joint allelic effect of NRPE1' and CMT2b' or CMT2a' on mCHH levels (Fig. 5A). mCHH levels on the RdDM-targeted TEs are primarily decided by NRPE1', and the effects of CMT2b' are insignificant (t-test p-value=0.51 at center of TE region). The effect is similar for the cmt2 knock-out. On the other hand, NRPE1'additively suppresses mCHH levels of CMT2-targeted TEs, so that CMT2b'/NRPE1'(found in two lines: Lag1-5 and Bran-1) showed a 20% reduction of average mCHH levels relative to  $NRPE1'_{ref}/CMT2_{ref}$ . Although the genome-wide phenotypic variation explained by NRPE1' was not large (0.8%; see Table S3), mCHH levels of CMT2-targeted TEs are well predicted by both loci (S9 Fig, S3 Table). The role of the RdDM pathway on the establishment of DNA methylation in CMT2-targeted TEs has been studied [4], and it appears to work on the edges of long TEs only (as shown in cmt2; see Fig. 5A). In contrast, the effect of the natural allelic variation at NRPE1'allele was observed over the entire TE, including the body. This suggests a qualitative difference between the natural alleles and the knock-out allele.

In summary, genotypes of NRPE1 and CMT2 generate further diversity of mCHH status over the genome. Given that both loci affect the pattern of methylation on CMT2-targeted TEs, it is worth noting that the allele frequencies at these two loci are strongly correlated. In particular, the genotype NRPE1'/CMT2b', which maximally

suppresses mCHH levels is only found in 2 of 1135 lines — an order of magnitude fewer than expected under random mating, and significantly rare compared to genome-wide SNPs of identical frequency (Figs 5B and S10; *p*-value < 0.01). This suggests selection against this combination, perhaps to avoid genome-wide hypomethylation. 168



Figure 5. CMT2b', and CMT2a' on mCHH levels in RdDM and CMT2-targeted TEs. (A) mCHH levels of TEs for six genotypes (left) and *nrpe1-11* and *cmt2* (right). 5', TE body, and 3' regions were divided into 20 sliding bins for CMT2- and RdDM-targeted TEs. (B) Allele frequencies of combinational genotypes between CMT2', and NRPE1' in 1135 lines. NRPE1' + ' and '-' indicate reference and alternative alleles. Five lines carrying CMT2b'/CMT2a' were omitted.

# NRPE1 and CMT2 alleles shape the longitudinal mCHH pattern

Previous studies have shown correlations of DNA methylation levels with several climate variables [10–12], but the genetic basis for this remains unclear. We examined whether the alleles at NRPE1 and CMT2 generate geographic patterns of mCHH levels 173 (Fig. 6). Variation at both loci show strong longitudinals patterns (NRPE1' r<sup>2</sup>=0.37, 174 p-value<2e-16; CMT2b' r<sup>2</sup>=0.02, p-value=4.4e-05; CMT2a' r<sup>2</sup>=0.006, p-value=0.03). 175 At NRPE1, the alternative allele is essentially only found in the east, and this is the 176 cause of a longitudinal cline in mCHH methylation on NRPE1-targeted TEs (r<sup>2</sup>=0.024, 177 p-value=3.0e-05 vs r<sup>2</sup>=0.002, p-value=0.26 after regressing out NRPE1') even after 178 correcting population structure (S11 Fig). 179

At *CMT2*, both alternative alleles are limited to Europe, where they appear 100 intermingled, but this causes no longitudinal cline for mCHH on CMT2-targeted TEs as 101 the alleles have opposite effects relative to the reference allele ( $r^2=0.001$ ; *p*-value=0.39; 102 Figs 6, S11). The distribution of *NRPE1*' alleles, which also affect CMT2-targeted TEs 103 contributes to the lack of a longitudinal pattern (*p*-value=0.03), consistent with the 104 observation above that selection may be acting to stabilize methylation. 105

170 171 172

bioRxiv preprint doi: https://doi.org/10.1101/819516; this version posted October 25, 2019. The copyright holder for this preprint (which was not certified by peer review) is the author/funder, who has granted bioRxiv a license to display the preprint in perpetuity. It is made available under aCC-BY 4.0 International license.



Figure 6. Geographical distribution of *NRPE1* and *CMT2* alleles, and longitudinal mCHH variation. Maps on the left show the distribution of *NRPE1'*, *CMT2b'*, and *CMT2a'* alleles, and the frequency of non-reference alleles along to longitude. Plots on the right show average mCHH levels of *NRPE1*- and *CMT2*-targeted TEs as a function of longitude. mCHH levels are average of *NRPE1'* and *CMT2b'*-targeted TEs. Colors of regression lines correspond to alleles; the black lines correspond to all lines.

# Discussion

In this paper we re-analyze the 1001 Epigenomes [12], focusing on mCHH patterns on individual TEs and TE families rather than on genome-wide averages performed in previous studies [10–12]. The advantages of this approach are evident. First, we were able to identify the well-known RdDM and CMT2 pathways using only natural variation data. We also identify several new associations, presumably corresponding to previously unknown members of these extensively-studied pathways (Figs 1-2). Second, the use of more fine-grained phenotypes allowed to refine previous associations, identifying candidate causal polymorphisms in both CMT2 and NRPE1 (Fig. 2). Furthermore, by comparing the genome-wide mCHH pattern with published data for loss-of-function mutations [13, 19], we were able to establish the causal involvement of these genes (Figs 3-4).

In terms of molecular mechanisms, our results largely confirm and complement previous studies [4,5,20]. The natural alleles of *CMT2* and *NRPE1* functionally behave much like loss-of-function alleles, albeit with some interesting differences that deserve further study. It is worth emphasizing in this context that these natural alleles have large effects, and are amenable to experimental studies. Perhaps because we are dealing with functional alleles, perhaps because we average over hundreds of lines, we get very clear pictures of which TEs are targeted by which *de novo* pathway (Figs 1 and S2). The mechanism underlying this targeting and the transition between pathways still remains unclear despite considerable effort.

Analysis of active TEs might be informative from this point of view. The current

186

187

188

189

190

191

192

193

194

195

196

197

198

199

200

201

202

203

204

205

206

study is limited to TEs annotated in the reference genome, and present at high frequency in the population. New TE insertions are likely to generate DNA methylation diversity [21] but analysis of this will have to await long-read genome sequencing of many lines, which will let us capture rare insertions, and study *de novo* silencing. [12, 22, 23].

Finally, we confirm the existence of major *trans*-acting polymorphisms affecting *de novo* DNA methylation [11, 12]. Based on currently available GWAS results, a genetic architecture characterized small numbers of genes of large effect is highly unusual, and is typically associated with adaptive polymorphism [24], but we can only speculate about what the adaptive value of variation in TE methylation would be. However, the idea of trade-offs and arms-races in a "genomic immune system" is not ridiculous such mechanisms clearly maintain polymorphism in other defense systems [25]. The geographic pattern observed here, with linkage disequilibrium between unlinked loci (Fig. 6), is certainly suggestive of selection.

# Materials and Methods

# Methylation data

Bisulfite sequencing data, leaves of plants grown under ambient conditions at SALK, published in the 1001 epigenome project was mapped on each pseudogenome from the 1001 genome project [12, 26], using a Methylpy pipeline

(https://bitbucket.org/schultzmattd/methylpy/wiki/Home). Methylation levels were calculated as weighted methylation levels [8]. TE regions were defined based on Col-0 by TAIR10 annotation, and 9,228 TEs having mapped reads in the region in all lines (n=774) were used for all analyses as common TEs. The CMT2 and the RdDM-targeted TEs were defined as it having DML (>0.1) between wild-type and drm1drm2 or cmt2 in Col-0 [13] as previously described [12]. The classification of TE families and superfamilies was based on TAIR10 [27].

# Statistical analysis

# Clustering

Clustering of TE families was conducted based on average mCHH levels across 774 lines (Fig. 1). The values were transformed into rank order per line and analyzed by hclust function with R (https://www.r-project.org/), with the agglomerative method 'complete'. All other clustering analyses were conducted with raw values as described in results using hclust function with default settings.

# GWAS

For GWAS of individual TEs and TE families, mCHH levels were transformed into rank 242 order across lines. Average mCHH in TE families were calculated for it of common TEs. 243 For GWAS of gene expression, 665 lines published in a part of the 1001 epigenome 244 project were used [12]. We obtained normalized gene expression values using fragments 245 per kilobase exon per million reads (FPKM) values published in Gene Expression 246 Omnibus (GSE80744) and transformed it into the most normal by Box cox method. 247 GWAS was performed using a linear mixed model [28,29] by LIMIX [30] with a full 248 genome SNP matrix from the 1001 genome project (10,709.949 SNPs), and population 249 structure was corrected by IBS matrix. A linear model without correction of population 250 structure was conducted using lm function in R (https://www.r-project.org). SNPs that 251 satisfied minor allele frequency (MAF) > 5% were used for association studies. 252

208

209

210

211

212

213

214

215

216

217

218

219

220

221

222

223

224

225

226

227

228

229

230

231

232

233

234

235

236

237

238

239

240

#### Meta-analysis

253

256

257

258

259

260

261

262

263

264

265

266

267

268

269

270

271

272

273

274

275

276

277

278

279

280

286

To combine p-values for each SNP calculated by GWAS, we used Fisher's methods as the following formula [31]. 254

$$X^{2} = -2\sum_{i=1}^{k} log(p_{i})$$
(1)

where  $p_i$  is *p*-value for ith GWAS, and *k* is the number of GWAS in the meta-analysis.  $X^2$  follows  $X^2$  distribution with 2k degrees of freedom. To optimize the threshold, we calculated false discovery rate (FDR) using the enrichment test with *a priori* gene list of 79 epigenetic regulators as described in [12]. The most significant *p*-value within 15 kb of a gene (MAF > 5%) was assigned as the significance of the gene.

LD  $(r^2)$  were calculated between all pairs of SNPs satisfied with the FDR threshold to determine independent GWAS peaks. In the case that a SNP pair has high LD  $(r^2>0.2)$ , a SNP having lower  $X^2$  scores were excluded from the list.

#### Correlation of the allelic effects and molecular phenotypes

Differential mCHH levels (DML) induced by alleles were estimated as differential average methylation levels between lines carrying the reference (Col-0) and the alternative allele for each TE. DML induced by mutants was calculated by the same way between wild-type and 86 loss-of-function mutants (Fig. 4; GSE39901; [13]) and *nrpe1* mutants (GSE93558 [19]). Spearman's correlation coefficients were calculated between DML for natural alleles, and mutants and empirical *p*-values were estimated using permutation test with 1500 randomly picked up SNPs along to genome (S4 Fig).

#### LD estimation

D' as standardized linkage disequilibrium was calculated as D'=D/D<sub>max</sub> [32]. D' was calculated between the target SNPs (*NRPE1*', *CMT2b*', and *CMT2a*') and genome-wide (unlinked) SNPs with same allele frequency of the target SNP. For example, *NRPE1*' (chr2: 16719071, MAF 9.0%) versus *CMT2b*' (chr4: 10417744, MAF 23.7%) was calculated between *NRPE1*' and all SNPs having the same MAF with *CMT2b*' (23.7%) on chromosome 1 and 3-5. The empirical *p*-value of observing an association was calculated using Fisher's exact test (one-sided).

#### Analysis of geographic patterns

Average mCHH levels of NRPE1- and CMT2-targeted TEs were calculated using TEs identified by GWAS ( $-\log_{10}p$ -value >=6 for NRPE1' and CMT2b'). Correlation between longitude and mCHH was calculated by a linear regression model for 728 lines ranging from longitude -25 to 100 in the 1001 epigenome project data (only SALK leaf samples). 283

# Acknowledgments

We thank Dr. Frederic Berger and Dr. Arturo Marí Ordóñez for critical reading of the manuscript, and Rahul Pisupati and Ümit Seren for technical support of data analyses (Gregor Mendel Institute of Molecular Plant Biology). 289

# References

1.	Law JA, Jacobsen SE. Establishing, maintaining and modifying DNA methylation patterns in plants and animals. Nat Rev Genet. 2010 Mar;11(3):204–220.	291 292
2.	Kawashima T, Berger F. Epigenetic reprogramming in plant sexual reproduction. Nat Rev Genet. 2014 Sep;15(9):613–624.	293 294
3.	Zhang H, Lang Z, Zhu JK. Dynamics and function of DNA methylation in plants. Nat Rev Mol Cell Biol. 2018 Aug;19(8):489–506.	295 296
4.	Zemach A, Kim MY, Hsieh PH, Coleman-Derr D, Eshed-Williams L, Thao K, et al. The Arabidopsis nucleosome remodeler DDM1 allows DNA methyltransferases to access H1-containing heterochromatin. Cell. 2013 Mar;153(1):193–205.	297 298 299 300
5.	Stroud H, Do T, Du J, Zhong X, Feng S, Johnson L, et al. Non-CG methylation patterns shape the epigenetic landscape in Arabidopsis. Nat Struct Mol Biol. 2014 Jan;21(1):64–72.	301 302 303
6.	Wassenegger M, Heimes S, Riedel L, Sänger HL. RNA-directed de novo methylation of genomic sequences in plants. Cell. 1994 Feb;76(3):567–576.	304 305
7.	Matzke MA, Mosher RA. RNA-directed DNA methylation: an epigenetic pathway of increasing complexity. Nat Rev Genet. 2014 Jun;15(6):394–408.	306 307
8.	Schmitz RJ, Ecker JR. Epigenetic and epigenomic variation in Arabidopsis thaliana. Trends Plant Sci. 2012 Mar;17(3):149–154.	308 309
9.	Weigel D, Colot V. Epialleles in plant evolution. Genome Biol. 2012 Oct;13(10):249.	310 311
10.	Shen X, De Jonge J, Forsberg SKG, Pettersson ME, Sheng Z, Hennig L, et al. Natural CMT2 variation is associated with genome-wide methylation changes and temperature seasonality. PLoS Genet. 2014 Dec;10(12):e1004842.	312 313 314
11.	Dubin MJ, Zhang P, Meng D, Remigereau MS, Osborne EJ, Paolo Casale F, et al. DNA methylation in Arabidopsis has a genetic basis and shows evidence of local adaptation. Elife. 2015 May;4:e05255.	315 316 317
12.	Kawakatsu T, Huang SSC, Jupe F, Sasaki E, Schmitz RJ, Urich MA, et al. Epigenomic Diversity in a Global Collection of Arabidopsis thaliana Accessions. Cell. 2016 Jul;166(2):492–505.	318 319 320
13.	Stroud H, Greenberg MVC, Feng S, Bernatavichute YV, Jacobsen SE. Comprehensive analysis of silencing mutants reveals complex regulation of the Arabidopsis methylome. Cell. 2013 Jan;152(1-2):352–364.	321 322 323
14.	Mi S, Cai T, Hu Y, Chen Y, Hodges E, Ni F, et al. Sorting of small RNAs into Arabidopsis argonaute complexes is directed by the 5' terminal nucleotide. Cell. 2008 Apr;133(1):116–127.	324 325 326
15.	Harris CJ, Scheibe M, Wongpalee SP, Liu W, Cornett EM, Vaughan RM, et al. A DNA methylation reader complex that enhances gene transcription. Science. 2018 Dec;362(6419):1182–1186.	327 328 329
16.	Goodstein DM, Shu S, Howson R, Neupane R, Hayes RD, Fazo J, et al. Phytozome: a comparative platform for green plant genomics;.	330 331

17	. Ogura T, Busch W. From phenotypes to causal sequences: using genome wide association studies to dissect the sequence basis for variation of plant development. Curr Opin Plant Biol. 2015 Feb;23:98–108.	332 333 334
18	. Gallagher MD, Chen-Plotkin AS. The Post-GWAS Era: From Association to Function. Am J Hum Genet. 2018 May;102(5):717–730.	335 336
19	. Wendte JM, Haag JR, Singh J, McKinlay A, Pontes OM, Pikaard CS. Functional Dissection of the Pol V Largest Subunit CTD in RNA-Directed DNA Methylation. Cell Rep. 2017 Jun;19(13):2796–2808.	337 338 339
20	. Zhong X, Hale CJ, Law JA, Johnson LM, Feng S, Tu A, et al. DDR complex facilitates global association of RNA polymerase V to promoters and evolutionarily young transposons. Nat Struct Mol Biol. 2012 Sep;19(9):870–875.	340 341 342
21	. Pignatta D, Erdmann RM, Scheer E, Picard CL, Bell GW, Gehring M. Natural epigenetic polymorphisms lead to intraspecific variation in Arabidopsis gene imprinting. Elife. 2014 Jul;3:e03198.	343 344 345
22	. Zapata L, Ding J, Willing EM, Hartwig B, Bezdan D, Jiao WB, et al. Chromosome-level assembly of Arabidopsis thaliana Ler reveals the extent of translocation and inversion polymorphisms. Proc Natl Acad Sci U S A. 2016 Jul;113(28):E4052–60.	346 347 348 349
23	. Pucker B, Holtgräwe D, Stadermann KB, Frey K, Huettel B, Reinhardt R, et al. A Chromosome-level Sequence Assembly Reveals the Structure of the Arabidopsis thaliana Nd-1 Genome and its Gene Set; 2018.	350 351 352
24	. Atwell S, Huang YS, Vilhjálmsson BJ, Willems G, Horton M, Li Y, et al. Genome-wide association study of 107 phenotypes in Arabidopsis thaliana inbred lines. Nature. 2010;465(7298):627–631.	353 354 355
25	. Todesco M, Balasubramanian S, Hu TT, Traw MB, Horton M, Epple P, et al. Natural allelic variation underlying a major fitness trade-off in Arabidopsis thaliana. Nature. 2010 Jun;465(7298):632–636.	356 357 358
26	. 1001 Genomes Consortium. 1,135 Genomes Reveal the Global Pattern of Polymorphism in Arabidopsis thaliana. Cell. 2016 Jul;166(2):481–491.	359 360
27	. Berardini TZ, Reiser L, Li D, Mezheritsky Y, Muller R, Strait E, et al. The Arabidopsis information resource: making and mining the "gold standard" annotated reference plant genome. Genesis. 2015;53(8):474–485.	361 362 363
28	. Yu J, Pressoir G, Briggs WH, Vroh Bi I, Yamasaki M, Doebley JF, et al. A unified mixed-model method for association mapping that accounts for multiple levels of relatedness. Nat Genet. 2006 Feb;38(2):203–208.	364 365 366
29	. Kang HM, Zaitlen NA, Wade CM, Kirby A, Heckerman D, Daly MJ, et al. Efficient control of population structure in model organism association mapping. Genetics. 2008 Mar;178(3):1709–1723.	367 368 369
30	. Lippert C, Casale FP, Rakitsch B, Stegle O. LIMIX: genetic analysis of multiple traits; 2014.	370 371
31	. Evangelou E, Ioannidis JPA. Meta-analysis methods for genome-wide association studies and beyond. Nat Rev Genet. 2013 Jun;14(6):379–389.	372 373
32	. Lewontin RC. The interaction of selection and linkage. I. General considerations; heterotic models. Genetics. 1964;.	374 375

bioRxiv preprint doi: https://doi.org/10.1101/819516; this version posted October 25, 2019. The copyright holder for this preprint (which was not certified by peer review) is the author/funder, who has granted bioRxiv a license to display the preprint in perpetuity. It is made available under aCC-BY 4.0 International license.

# Supporting Information

#### S1 Fig

Enrichment of *a priori* DNA methylation responsible genes in meta-analysis. Enrichment and FDR 20% based on *a priori* genes (see Methods and also [12]). The horizontal dashed line at 0.2 corresponds to FDR 20%.

### S2 Fig

Effects of TE length and the location on target specificity of NRPE1 and CMT2. Bar plots indicate the average length of TE families ordered by the length with GWAS *p*-values for three alleles (line plots; see also S1 Table) and the proportion of TEs located around centromeric regions (black fraction in bar plots; 1Mbp from centromeric regions).

#### S3 Fig

Effect of population structure on GWAS results. (A) Scatter plots show correlations of differential mCHH levels (DML) induced by alleles and mutants for each TE. DML for alleles was estimated as average differences of mCHH levels between lines carrying reference and non-reference alleles, whereas for mutants it was estimated between wild-type and nrpe1-11 or cmt2. Colors of dots in the scatter plots show the significance of the allelic effects as  $-\log_{10}p$ -value in GWAS (a linear model without correction of population structure). Density plots on Y and X-axis show distributions of the allelic effects for TEs. (B) Effects of population structure for mCHH levels of individual TEs. Scatter plots show  $-\log_{10}p$ -values estimated by a linear model (lm in X-axis) and a linear-mixed model (lmm in Y-axis).

# S4 Fig

**Permutation tests for allelic effects.** Spearman's correlation coefficients (r) were calculated between DML of candidate mutants (*nrpe1-11* and *cmt2*) and 1500 randomly picked up SNPs over the genome (see Methods). Orange arrows show r of *NRPE1'*, *CMT2a'*, and *CMT2b'*. All allelic effects were significantly stronger than randomly picked up SNPs (p < 0.001).

# S5 Fig

LD effects on the correlations between allelic effects and mutant phenotypes. Each dot shows the absolute value of Spearman's correlation coefficients r between DML of the three alleles and 67 single knockout mutants (Stroud et al., 2013) along with the gene location on the genome.

# S6 Fig

**GWAS for NRPE1 expression.** Manhattan plots and the *cis* peaks for NRPE1 expression (n=665; leaf tissue under  $21^{\circ}$ C). Horizontal lines show the threshold (*p*-value 5% Bonferroni correction).

# S7 Fig

The allelic effects of AGO1 in the RdDM pathway and the similarity to AGO1 activity. Scatter plots show correlations of DML induced by NRPE1' and

376

377

378

379

380

381

382

383

385

386

387

388

389

390

391

392

393

394

395

396

397

398

399

400

401

402

403

404

405

406

407

409

410

411

412

413

414

mutants, nrpe1-11 and ago1, for each TE. DML for alleles was estimated as average differences of mCHH levels between lines carrying reference and non-reference alleles, whereas it for mutants was estimated between wild-type and *nrpe1-11* and *aq01* loss-of-function. Colors of dots in the scatter plots show the significance of the allelic effects as  $-\log_{10} p$ -value in GWAS. Density plots on Y and X-axis show distributions of the allelic effects for TEs.

# S8 Fig

Target specificities of the allelic effects of NRPE1', CMT2b', and CMT2a' on mCHH levels of individual TEs. (A) Compositions of TE-superfamilies identified by GWAS, population-based average, or loss-of-function mutants at 0 to 90 percentile thresholds. (B) The scatter plot shows the correlation between DML induced by NRPE1' and nrpe1 loss-of-function with 95% confident prediction. Blue dots indicate TEs showing nrpe1-1 loss-of-function specific effects on DML, and red dots indicate TEs that were not detected by GWAS regardless of the DML ( $lm - log_{10}p$ -value > 3). (C) Composition of TE-superfamilies shown in panel B (blue and red dots).

# S9 Fig

Allelic effects between RdDM and CMT2 pathways. Correlation between molecular phenotypes of nrpe1-11 and cmt2 and the allelic effects on mCHH levels of TEs. NRPE1, CMT2-targeted, and untargeted TEs are shown in blue, red, and grey respectively based on GWAS results  $(-\log_{10}p\text{-value}>6 \text{ for } NRPE1' \text{ and } CMT2b')$ . Regression lines are corresponding to NRPE1 and CMT2-targeted TEs.

# S10 Fig

Genome-wide pattern of LD for the NRPE1 and CMT2 alleles. Plot A compares the value of D' between NRPE1' and CMT2b' (orange arrow) to the distribution of D' between NRPE1' and genome-wide (unlinked) SNP of the same frequency as CMT2b' on the left. The plot on the right shows the corresponding distribution of *p*-values calculated using Fisher's Exact Test (one-sided). The empirical *p*-value of observing an association this strong is less 0.01. Plots B and C show the same, focusing on CMT2a' and NRPE1', and CMT2b' and NRPE1', respectively.

# S11 Fig

9

Allelic effects on the geographical cline of mCHH levels. Plots show average 446 mCHH levels of NRPE1- and CMT2-targeted TEs by taking into account population 447 structure (BLUP) as a function of longitude. mCHH levels are averages of NRPE1' and 448 CMT2b'-targeted TEs. Colors of regression lines correspond to alleles; the black lines correspond to all lines.

S1 Table	451
GWAS results for average mCHH of TE families	452
S2 Table	453
Top SNPs associated with mCHH variation (FDR20)	454

# 422 423

416

417

418

419

420

421

427 428

429

430

431

432

424

425

433 434 435

437

440 441

438

442 443 444

S3 Table	455
Genetic effects on mCHH variation	456
S4 Table Compositions of TE superfamilies in Col-0 reference and it of common the population (n=774)	457 in 458 459